University of Toronto, Department of Electrical and Computer Engineering

ECE241F - Digital Systems - Basic Cour se Information

Fall 1997 S. Brown, J. Rose Instructor and Lecture Inf ormation

Grading

Midterm

Is on Thursday October 9th, 2-3pm, Room TBA. This is an early mid-term! Closed book. Exam

Type D - examiner specified aids, and no calculators of any kind permitted. Text

A draft of a new text written by Brown and Vranesic available in SF B540, at low cost. Software

CAD software that runs under Windows 95/Windows NT will be available free, courtesy of Altera Corp. You will need a Windows machine with at least 32MB of memory. This software is also available on the eecg ugsparcs system and on Windows machines in the lab.

On-Line Documents

http://www.eecg.toronto.edu/~jayar/courses/ece241F/home.html

Section L0101 L0201 Instructor: Jonathan Rose Stephen Brown Office: LP 484E SF 2001 Phone: 978-6992 978-1647 Email: jayar@eecg.toronto.edu brown@eecg.toronto.edu Home Page: www.eecg.toronto.edu/~jayar www.eecg.toronto.edu/~brown Office Hours Please contact by email or during lectures to make appointment

Lecture Rooms

Tues: WB 116 SF 1105 Thurs SF 1105 MC 252 Fri: WB 116 MC 252

Labs 10% Midterm 25% Project 10% Exam 55%